Author's response to reviews

Title: Preventing childhood obesity during infancy in UK primary care: a mixed-methods study of HCPs' knowledge, beliefs and practice

Authors:

Sarah A Redsell (Sarah.Redsell@nottingham.ac.uk)
Philippa J Atkinson (p.j.atkinson@bopenworld.com)
Dilip Nathan (dilip.nathan@nhs.net)
A Niroshan Siriwardena (nsiriwardena@lincoln.ac.uk)
Judy A Swift (judy.swift@nottingham.ac.uk)
Cristine Glazebrook (cris.glazebrook@nottingham.ac.uk)

Version: 3 Date: 14 June 2011

Author's response to reviews:

Dear Eden

Thank you for accepting this paper for publication. I have moved the two sections "Data sharing" and "Body giving ethical approval" from the end of the manuscript into the Methods section of the manuscript.

I have not made any further changes. The paper I am submitting to you still has the "track changes" function on. If you would prefer a different version please let me know.

Best wishes

Sarah